Title: A blended knowledge translation initiative to improve colorectal cancer staging [ISRCTN56824239]

Authors:

Frances C. Wright (frances.wright@sw.ca)
Calvin H.L. Law (calvin.law@sw.ca)
Linda D. Last (linda.last@sw.ca)
Neil Klar (neil.klar@cancercare.on.ca)
David P. Ryan (david.ryan@sw.ca)
Andrew J. Smith (andy.smith@sw.ca)

Version: 8 Date: 10 November 2005

Author's response to reviews:

Changes were made as per suggestions from the reviewer. However, we were only able to identify one minor error in the references (not two) and this has now been corrected (correction of spelling of author's name "Locock").